Author’s response to reviews

Title: Patients’ Perspectives of Outcomes after Total Knee and Total Hip Arthroplasty: A Nominal Group Study

Authors:

Susan Goodman (GoodmanS@HSS.EDU)
Bella Mehta (mehtab@hss.edu)
Serene Mirza (mirzas@hss.edu)
Mark Figgie (figgiem@hss.edu)
Michael Alexiades (alexiadesm@hss.edu)
Jose Rodriguez (RodriguezJose@hss.edu)
Peter Sculco (sculcoP@hss.edu)
Michael Parks (ParksM@hss.edu)
Jasvinder Singh (jasvinder.md@gmail.com)

Version: 4 Date: 17 Oct 2019

Author’s response to reviews:

1. Formatting

Please include a Conclusion section in your main manuscript.

Response: Conclusion section included on page 15 after discussion.

2. Ethics approval and consent to participate

Please include your consent to participate statement in your 'ethics approval and consent to participate' section.

Response: IRB approval and consent to participate information included in the ‘ethics approval and consent to participate’ section under the Declarations on page 18.

3. Authors’ Contributions
-- Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

-- Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the final manuscript, and ensure that this is the case.

Response: Author contributions, indicated by full initials, and a statement declaring all authors have read and approved the final manuscript are included in the Author Contributions section under the Declarations on page 18.

4. Acknowledgements

We note that you have not included an acknowledgements section in your Declarations section. If you have no acknowledgements please put ‘Not Applicable’ in this section.

Response: We do not have any acknowledgements. An acknowledgements section and statement of “Not Applicable” has been added under the Declarations on page 18.

5. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: A final clean manuscript is included in the submission of this revision. All figures, tables, and supplementary files are properly cited within the text.